Author’s response to reviews

Title: Home based records for poor mothers and children in Afghanistan, a cross sectional population based study

Authors:
Sayed Saeedzai (atasayedzai@gmail.com)
Iftekhar Sadaat (iftekharsadaat@gmail.com)
Zelaikha Anwari (zanwari.mophrhd@gmail.com)
Shafiqullah Hemat (shafiqullahhemat8@gmail.com)
Shakir Hadad (hadad.rhd@gmail.com)
Keiko Osaki (Osaki.Keiko@jica.go.jp)
Megumi Asaba (masaba@tulane.edu)
Yohei Ishiguro (Ishiguro.Yohei@jica.go.jp)
Rasuli Mudassir (RasuliAhmadmuddassir.AN@jica.go.jp)
Jane Burke (Machlin.burke@gmail.com)
Ariel Higgins-Steele (ahiggins@unicef.org)
Khaksar Yousufi (kyousufi@unicef.org)
Karen Margaret Edmond (karen.edmond@uwa.edu.au)
Version: 4 Date: 26 May 2019

Author’s response to reviews:

Dear BMC Public Health,

Re: PUBH-D-18-03825R4. Home based records for poor mothers and children in Afghanistan, a cross sectional population based study Sayed Saeedzai; Iftekhar Sadaat; Zelaikha Anwari; Yohei Ishiguro; Shafiqullah Hemat; Shakir Hadad; Keiko Osaki; Megumi Asaba; Rasuli Mudassir; Jane Burke; Ariel Higgins-Steele; Khaksar Yousufi; Karen Margaret Edmond

Thank you for your feedback. We respond point by point below and have uploaded the revised manuscript. We look forward to your feedback. Best regards, Karen Edmond
1. Please modify the author list in your title page to reflect the proposed list in the Authorship Change Form.

This has now been done and the revised manuscript is attached